Author’s response to reviews

Title: Using eHealth to engage and retain priority populations in the HIV treatment and care cascade in the Asia-Pacific region: a systematic review of literature

Authors:

Julianita Purnomo (lia.purnomo@health.nsw.gov.au)
Katherine Coote (katherine.coote@health.nsw.gov.au)
Limin Mao (limin.mao@unsw.edu.au)
Ling Fan (lfan16@yahoo.com.au)
Julian Gold (julian.gold@health.nsw.gov.au)
Raghib Ahmad (raghib.ahmad@health.nsw.gov.au)
Lei Zhang (lei.zhang1@monash.edu.au)

Version: 2 Date: 05 Dec 2017

Author’s response to reviews:

Mirjam Kretzschmar
BMC Infectious Diseases

https://bmcinfectdis.biomedcentral.com/

5 December 2017

Dear Mirjam Kretzschmar,

Re: Resubmission of manuscript “Using eHealth to engage and retain priority populations in the HIV treatment and care cascade in the Asia-Pacific region: a systematic review of literature” (INFD-D-17-00472R2)
Thank you for the opportunity to revise and resubmit a revised version of the manuscript to BMC Infectious Diseases.

We would like to thank the Editor and the Reviewer 1 for their comments. Your feedback has led to further improvement in the quality of our revised paper. Revisions in the manuscript are shown in red track changes.

We hope that the latest revisions in the manuscript and our accompanying responses will be sufficient to make our manuscript suitable for publication in the BMC Infectious Diseases.

We look forward to hearing from you regarding our resubmission. We would be glad to respond to any further questions and comments that you may have.

Yours sincerely,

Julianita Purnomo

on behalf of the authors

==================================================================

Technical Comments:

Editor Comments:

Please check the manuscript for language errors and update accordingly. A revised version of the manuscript can be accepted for publication.

Authors’ response:

We apologise for the language errors in the manuscript. We have fixed all typographical and grammatical errors and carefully proof-read the revised manuscript to eliminate all such errors.

Reviewer reports:

John De Wit (Reviewer 1): This is a timely literature review and evidence synthesis. I commend authors on the extension and revision they have undertaken. I did notice some language glitches and recommend careful final editing, of newly added text in particular.

Authors’ response:
We appreciate the positive feedback from the reviewer. We have endeavoured to address language glitches. We have carefully proofread the manuscript and caught a number of typographical and grammatical errors, which have been addressed. See track changes throughout.